Author's response to reviews

Title: Perspectives on gastroesophageal reflux disease in primary care: the REFLEX study of patient-physician agreement

Authors:

   Etienne Dorval (dorval@med.univ-tours.fr)
   Jean-Francois Rey (jean-francois.rey@wanadoo.fr)
   Christine Soufflet (Christine.Soufflet@astrazeneca.com)
   Katarina Halling (khalling@patientreported.com)
   Philippe Barthélemy (Philippe.Barthelemy@astrazeneca.com)

Version: 4 Date: 1 March 2011

Author's response to reviews: see over
Nina Titmus  
On behalf of Christna Chap, PhD  
Executive Editor in training  
BioMed Central  
236 Gray's Inn Road  
London, WC1X 8HB  
UK  

1 March 2011  

Dear Dr Chap,  

Please find below a point-by-point description of the formatting changes made to the final version of the manuscript as requested.  

**Major revisions.**  

*Affiliations:* the author affiliations have been provided in full.  

*Abstract:* the abstract headings have been modified to follow the specified structure.  
The information regarding previously published work has been moved under the abstract using the heading previous publication. The original citation has been given in full.  

**Minor revisions.**  

*Running head:* has been removed.  

*Funding:* statement has been moved to acknowledgement section.  

*Key words:* have been removed.  

The manuscript has also been proof-read and is regarded as being ready for publication.  

Yours sincerely,  

Etienne Dorval  
Hôpital Trousseau  
Tours 37044  
France  
Tel: +33 (0)2 47 47 59 16  
Fax: +33 (0)2 47 47 84 28  
E-mail: dorval@med.univ-tours.fr